Author's response to reviews

Title: Sample size considerations using mathematical models: an example with Chlamydia trachomatis infection and its sequelae pelvic inflammatory disease

Authors:

Sereina A Herzog (herzog.sereina@gmail.com)
Andrea Berghold (andrea.berghold@medunigraz.at)
Nicola Low (low@ispm.unibe.ch)

Version: 2 Date: 16 January 2015

Author's response to reviews:

Dear Editors,

We clarified in the method section that for the mathematical model no ethical aspects have to be considered as no research subjects are involved. We included an acknowledgment section.

Yours sincerely, on behalf of all co-authors,
Sereina Herzog, PhD